Author’s response to reviews

Title: Estimating the annual risk of infection with Mycobacterium tuberculosis among adolescents in Western Kenya in preparation for TB vaccine trials

Authors:
Videlis Nduba (vnduba@gmail.com)
Anna van’t Hoog (a.h.vanthoog@aighd.org)
Annefleur de Bruijn (annefleurdebruijn@hotmail.com)
Ellen Mitchell (mitchell@iss.nl)
Kayla Laserson (kayla.laserson@gatesfoundation.org)
Martien Borgdorff (m.w.borgdorff@amc.uva.nl)

Version: 2 Date: 16 May 2019

Author’s response to reviews:

I have included presentation of this results in a previous abstract in the acknowledgement section. I have also revised the abstract results section to be different form the previous abstract presented in the Nairobi Hospital conference.